

Article

Individual Investors' Learning Behavior and Its Impact on Their Herd Bias: An Integrated Analysis in the Context of Stock Trading

Kalugala Vidanalage Aruna Shantha^{1,2}

- ¹ School of Management, Wuhan University of Technology, Wuhan 430070, China; shantharuna@yahoo.com or arunashantha@mgt.rjt.ac.lk
- ² Faculty of Management Studies, Rajarata University of Sri Lanka, Mihintale 50300, Sri Lanka

Received: 24 January 2019; Accepted: 4 March 2019; Published: 8 March 2019



MDF

Abstract: The efficient functioning of capital markets ensures that information on companies' sustainable development endeavors is fully and instantly incorporated into stock prices, which facilitates them in raising capital requirements at a lower cost. It, however, is impaired when market participants are inclined to behavioral biases. The Adaptive Market Hypothesis predicts that such behavioral biases are evolutionary. In that sense, market participants are capable of learning their behavioral mistakes and adapting to market conditions over time. Based on this perspective, this paper aims to explore how learning occurs within individual investors to reduce their herd bias. The data was collected by distributing a web-based self-administrated questionnaire to a sample of 1000 individual investors of the Colombo Stock Exchange, who were randomly selected during a period from March to August 2018. A total of 189 responses were received, which were analyzed using the structural equation modelling technique to test the hypotheses of the theoretical model. The results show that learning takes place when investors cognitively evaluate past trading experiences, which is induced by their desire for learning, and, consequently, reduces their herd bias. However, as the model predicts, strengthening this cognitive reflection from the relationship with the investment advisor and social learning among investors through their peer-relationships appear to be absent due to uncertain market conditions prevailed during the study period and dominance of unsophisticated investors in the market. From these findings, this paper concludes that the cognitive reflection of past experiences and the nature of the trading environment determine the extent of learning within individual investors.

Keywords: herding; behavioral bias; adaptive market hypothesis; self-reflection; Colombo stock exchange; market efficiency; investor education

"Do not be embarrassed by your failures, learn from them and start again."

Richard Branson

1. Introduction

The concept of sustainable development has now become an integral aspect of a company's strategic planning process. Therefore, it has become a guiding principle when formulating business policies and strategies in striving to achieve its objectives. As stated in Brundtland [1], "sustainable development implies meeting the needs of the present without compromising the ability of future generations to meet their own needs." When concerning the financing side, capital markets should promote sustainable development by facilitating companies to raise capital at a lower cost to finance their endeavor to become sustainable. However, Waygood [2] highlights that this role is currently impaired due to inefficiencies of capital markets which, from the sustainable development viewpoint,

indicate the inability of the markets to recognize and reward companies' right conducts to become sustainable. As a result, the information on companies' sustainable development endeavor is not fully and instantly incorporated into prices of their stocks. Waygood [2] further emphasizes that the failure of the investors' predictive power is the main reason of market inefficiency.

The behavioral finance literature shows that individual investors are less sophisticated than institutional investors, primarily due to their limited attention, memory, time, education, and processing capabilities. As a consequence, they are more likely to use simple heuristics or rules of thumb in their decision making, which would become maladaptive in a dynamic market environment [3–5]. The literature suggests that such maladaptive heuristics indicate irrationality of investors, which is classified as behavioral biases or mistakes in their decision-making [6]. These behavioral biases could give rise to deviation of prices of securities from their fundamental values, which results in inefficiencies in financial markets. Accordingly, investor sophistication plays a significant role in efficient functioning of a capital market so that capital is allocated to companies' sustainable developments at a lower cost, which enhances their long-term shareholder value.

One of behavioral biases extensively discussed in the literature is herding. It is one's propensity to abandon information and belief, and infer them from the actions of others in making choices. Despite the evidence on herd bias under different market states and characteristics, and consequences such as market bubbles, crashes, and increased volatility, it is interesting to note that certain previous studies reveal its declining tendency over time [7,8]. This implies a movement towards the market efficiency status. However, to the best of my knowledge, there is no study in the literature providing empirical evidence on the factors that ground for the diminishing herd behavior in a financial market. Accordingly, this paper intends to fill this gap in the literature by studying a case of a stock market where herd bias appears to decrease over time.

Given the research gap this work addresses, the Colombo Stock Exchange (CSE) of Sri Lanka, a frontier stock market, appears to be an ideal case to study due to the following reasons. First, when compared to developed and emerging markets, frontier markets are more vulnerable for herding due to higher uncertainty in an informational and trading environment [9]. Second, political uncertainty and economic crisis of the country during the past few years, and weaknesses in the regulatory structure of the CSE would have further intensified the market uncertainties, stimulating investors to herd when trading stocks. Third, the CSE exhibits strong herd bias during the 2000–2012 period, which, tends to decline and disappears afterwards [8,10].

The effectiveness in regulatory reforms and investor learning behavior are the two possible reasons suggested in the literature for the decline in herd behavior [7,8]. The CSE, however, has not undergone significant regulatory reforms during the period over which the anti-herding occurred. Therefore, this study aims to explore how learning occurs within individual investors to minimize their herd bias. The data was collected through a web-based self-administrated questionnaire distributed to a sample of 1000 individual investors randomly selected during the period of March to August 2018. The responses received, totaling 189, were analyzed using the structural equation modelling technique to test the hypotheses. Supporting the herding literature, the results reveal that, in the presence of uncertain market conditions, herd bias tends to exaggerate among investors through the effect of their peer-relationships. Conversely, it is also evident that they incline to learn the irrationality of herding by reflecting on their past experiences. Hence, they shift away from such irrational behavior when trading stocks. As a consequence, consistent with the previous studies on the CSE, herd bias appears to decline at the aggregate market level since the magnitude of the investors' learning is large enough to outweigh the increased herd bias through their peer-relationships.

This study provides the following five contributions to the academia and industry. First, according to the herding literature, an examination of investors' learning behavior by integrating its impact on their behavioral biases has not been conducted in a single study at the individual unit of analysis. Hence, this is the first of its kind of studies providing empirical evidence on how learning arises within individual investors to reduce their herd tendency occurring when trading stocks. The findings of this

study can be adopted when designing training programs for individual investors to improve their sophistication for lowering behavioral biases in decision-making. Second, this study is the first to apply the model of investor learning behavior proposed by Shantha et al. [11], which facilitates the examination of the learning behavior of the individual investors based on the data collected in a real market setting. Third, it attempts to extend the works of Shantha [8] and Xiaofang and Shantha [10] by exploring the factors that account for the declining herd behavior in the CSE. Fourth, the empirical results are derived from primary data sources, which is limitedly available in the behavioral finance literature [6]. Fifth, this study is conducted on a frontier stock market, which is a category of financial markets that, on one hand, is an ideal ground for the examination of herding and, on the other hand, has not had much focus in the existing herding studies so far [9,12].

The reminder of the paper is organized as follows. Section 2 elaborates how investor sophistication relates to the sustainable development of companies and provides a review of the relevant literature on herding and learning behaviors of investors. Section 3 discusses the nature of trading and informational environment of the CSE, which would have stimulated herd behavior in the market. Section 4 introduces the conceptual model and hypotheses. The research design and methodology used in this work are presented in Section 5. Section 6 shows the results related to the assessment of the reliability and validity of the model's constructs, while Section 7 provides the empirical test of the hypotheses to infer about the nature and extent of learning and its effect on herd bias. Section 8 concludes the paper with its implications to practice. The limitations of this study and the avenues for future research are discussed in Section 9.

2. Literature Review

2.1. Investor Sophistication and Sustainable Development

The capital market is important to a country since it provides finance for the economic development. It mobilizes savings of surplus units into medium to long-term investments in the forms of financial assets such as equity shares and corporate bonds, and thereby, enables companies to raise capital to finance their development projects. However, the cost of capital limits the amount of finance that can be raised for developments. In other words, a higher cost of capital would reduce the extent of activities that a company could undertake for its sustainable development. Hence, the sustainable development would be enhanced if the capital required for such long-term development plans could be raised at a lower cost.

The cost of capital is the average cost of financial assets that a company sources to obtain its long-term capital. Most of such financial assets, if not all, are traded on the capital market. Investors allot their funds by buying and selling of them, which ultimately affect the cost of the capital of the issuing companies. Hence, a capital market should facilitate investors to recognize the sustainability of development efforts of companies and allot funds efficiently so that the companies can raise capital at a lower cost. For this purpose, the markets should be allocationally efficient. The allocational efficiency arises when relevant information to judge risk-return and sustainability of development projects is readily available to investors to make their decisions. However, the information availability does not merely guarantee that investors recognize a sustainable endeavor and produce efficient allocation of their funds. Rather, the investors should possess the capacity to use such information to make accurate decisions. Accordingly, the investor sophistication plays a vital role in the efficient functioning of capital markets so that capital could be raised for sustainable developments at a lower cost [2].

2.2. Previous Studies on Herd Behavior

Herd behavior is one of the behavioral biases that has been extensively investigated in behavioral finance over the past few decades. The literature reveals the nature of herding, reasons underlying its occurrence, and its consequences for the functioning of markets [9,13–15]. The previous studies also find that herding is prevalent among different investor-types (for example, retail and institutional

investors) and likely to exaggerate through different market states and characteristics (for example, up vs. down market movement days, periods of high vs. low volatility, high vs. low trading volume days, cross-country effects, and effect of macro-economic factors). These studies are mostly conducted with respect to developed and emerging stock markets. However, compared to the developed and emerging markets, the frontier markets are weaker in terms of lower transparency, lower liquidity, higher information asymmetry, dominance of noise trading, and higher volatility. As a result, herding can be expected to be more prominent in frontier markets [9,12,16]. Thus, greater attention has focused on the study of herd phenomenon in frontier stock exchanges.

The evolutionary nature of behavioral biases is another line of research presently being considered in the behavioral finance. It emerged with the Adaptive Market Hypothesis (AMH) proposed by Lo [3–5], a theoretical framework which predicts an evolving efficiency of financial markets. According to the AMH, market participants' behavior is evolutionary over time and in response to market dynamics, and this evolutionary process enables them to learn their biased behaviors for adapting to market conditions. Hence, it can be expected that biases tend to decline and, as a result, the market may approach its efficiency status through their mistake-learning process. Supporting this prediction, Ito et al. [17] find that investors' learning behavior is an underlying factor for the evolving efficiency of financial markets.

The literature provides evidence for the evolving nature of herd behavior in financial markets. Choe et al. [18] and Hwang and Salmon [19] find a tendency to decrease herd bias after crisis periods. In the context of emerging markets, the study of Yao, Ma, and He [7] shows that herd behavior diminishes over time in the Chinese Stock Markets. Furthermore, relating to frontier stock markets, Nguyen [20] finds strong evidence of herding during the 2009–2016 period, whereas no evidence of this behavior afterwards was present in the stock market of Vietnam. Both Nguyen [20] and Yao, Ma and He [7] suggest the effectiveness of regulatory reforms in the respective markets as a possible reason for the decline in herd bias. Similar results have been reported with respect to the CSE of Sri Lanka. Shantha [8] and Xiaofang and Shantha [10] reveal a strong tendency to herd from 2000–2009 (a period of political uncertainty due to civil war) and 2009–2012 (market bubble and crash periods), and the evidence on anti-herding behavior after the market crash. Since there has not been significant regulatory reforms during this anti-herding period, Shantha [8] suggests that investors may have learned the irrationality of herding from the financial losses experienced during the period of the market crash, which reduced their herd tendency. The current study attempts to extend the work of Shantha [8] by exploring how the investors' learning has been effected to lower and/or disappear herd behavior in the CSE.

2.3. Previous Studies on Learning Behavior

The study of investors' learning has mostly been carried out in artificial market environments using agent-based financial models, postulating two approaches of learning that an investor engages in: individual learning and social learning [21–28]. The former represents an investor's attempt to learn by his/her own, whereas, in the latter case, the learning occurs by imitating the others' behaviors. Yamamoto [22] finds that wealthy investors involved in individual learning while other investors follow social learning by imitating the actions of the wealthy investors. Yamamoto [22] and Bossan, Jann and Hammerstein [27] show that social learning efforts are widespread among individual learning. Bossan, Jann and Hammerstein [27], by examining different social learning procedures, find that payoff-biased and imitating the wealthiest yield superior outcomes than individual learning.

When modeling the individual form of learning, the agent-based models typically assume the reinforcement learning (RL) since investors' heuristic biases would result in trial-and-error behaviors for adapting to dynamic market conditions. However, Pastore, Esposito, and Vasilaki [28] find that only a subset of players in their study sample follow the RL, which is insufficient to confirm the RL as a component of investors' trading decisions. Furthermore, if the RL assumption holds true, a higher

stock trading experience would result in better investment strategies. However, the previous studies provide mix evidence related to this prediction since, despite the evidence supporting this RL [29–37], certain studies show some contradictory evidence as follows. Chevalier and Ellison [38] find a negative relationship between the investment experience and performance. In addition, Agarwal et al. [39] reveal that the relationship between investment experience and performance takes a reverse U shape form, which represents the fact that investment performance tends to decline when the experience increases beyond a particular level. Bhandari and Deaves [40], Xiao [41], Bodnaruk and Simonov [42], Wulfmeyer [43], and Chang [44] also find that the experience escalates overconfidence and disposition effect, which results in a lower investment performance. On the other hand, when social learning is concerned, the herding literature reveals that imitating others' behaviors is an irrationality, which, in the event of becoming a market-wide trend, causes unfavorable effects to a financial market such as increased volatility, speculative bubbles, and crashes [6,15]. In addition to the way learning is assumed, the agent-based models provide a weak representation of investor behavior due to unrealistic parameter configurations adopted for formulating the models such as assumptions on the number of assets traded, number of agents, timing of decisions, information and execution of trade, and speed at which agents update their behavioral rules [24,45]. Furthermore, a large number of such parameter configurations poses difficulties in modeling learning processes, which affects the reasonableness of evidence generated by these models [46]. Hence, in view of these findings, the existing agent-based studies do not appear to deliver a fair representation of an investor's learning behavior occurred in a real stock market.

3. Regulatory and Trading Environment of the CSE

The CSE was established in 1985 as a company limited by guarantee under the Companies Act No. 17 of 1982 and mutually owned by 15 stockbrokers. It provides a trading platform for both debt and equity instruments and has 297 listed companies, representing 20 business sectors with a market capitalization of LKR 2839.45 billion as of 31 December, 2018, which corresponds to about 20 percent of the GDP of the country. A majority of stock trading is executed by local investors, which is approximately 85% of the total number of stock traded and 96% of the total number of trades. In terms of investor-types, retail investors currently dominate the market, accounting for more than 90% of the total market capitalization.

In 1987, the Securities Council Act No. 36 of 1987 was enacted, which established the regulator, the Securities and Exchange Commission (SEC) of Sri Lanka, to facilitate orderly and fair functioning of the capital market. The act was amended in 1991, 2003, and 2009 to strengthen the aspects related to regulation and supervision. However, during the last decade, weaknesses in the current regulatory structure are apparent with the market bubble, a crash that occurred over the 2009–2012 period, and certain instances of insider trading and price manipulations, which resulted in unfavorable consequences such as information asymmetry, low transparency, high volatility, and a few number of securities actively traded in the market. Particularly, over the past few years, political and economic crises of the country prompted an uncertain trading environment, causing impediments to the investor participation and trading activities of the CSE. Consequently, the market experienced a declining trend, as reflected by its All Share Price Index which, recorded at 7605.79 on the first day of the new government in 2015, dropped to 6052.37 as of 31 December, 2018, which is a decline of 20% during this four-year period.

This uncertain informational and trading environment of the CSE may motivate investors to suppress their own information and imitate others' behaviors for information when making their trading decisions [47]. On the other hand, consistent with the implications of Lo [4,5] and Shantha [8], such irrational behaviors of investors could decline when they learn based on their past experiences, which is further stimulated through the educational initiatives of the CSE. The awareness campaigns and investor education programs for investing in the stock market are regularly conducted by the CSE in association with the SEC to enhance their financial literacy and stock market participation.

In 2016, it launched an online educational portal to offer learning material, training and analyses, which added a new dimension to the investor education initiatives. Thus, it can be expected that the investors, in light of these educational initiatives and their trading experiences, would be able to learn their mistakes such as herding and shift away from such irrational behaviors when making decisions to buy/sell stock.

4. Conceptual Model and Hypotheses

The conceptual model used in this paper is based on the work of Shantha, Xiaofang, and Gamini [11], which, incorporating ideas from the behavioral finance and learning literature, proposes a behavioral model to examine investor learning behavior in the context of stock trading. Similar to the learning behavior modeled through agent-based models, their model assumes that investors exhibit individual learning and/or social learning for adapting their investment behavior to dynamic market conditions. However, this model is different from the existing agent-based models in terms of the following features. Unlike the agent-based models that configure only the parameters related to learning behavior of investors, the new behavioral model can be used to study the learning behavior by integrating its effects on their behavioral biases or frame of reference on investing. Furthermore, based on the learning literature, it conceptualizes learning holistically by incorporating cognitive, affective, and social aspects of learning as well as the behavioral aspects typically concerned in the existing agent-based models. The extent of individual and social learning conceptualized in this model can also be empirically measured using primary data obtained in a real market setting.

The model, as shown in Figure 1, assumes that the individual learning occurs within an investor when he/she cognitively evaluates past trading experiences. Accordingly, this cognitive evaluation of experiences, known as "self-reflection," is the mechanism of the individual form of learning, which, in turn, reduces herd bias, as given by the following hypotheses.

Hypothesis 1 (H1). *An investor's trading experience (TE) is positively related to the extent of self-reflection (SR) he/she has when learning.*

Hypothesis 2 (H2). *The level of SR is negatively related to the extent of herd bias (HERD) that occurred when trading stocks.*

Hypothesis 3 (H3). *SR mediates the relationship between TE and HERD that occurred when trading stocks.*

In addition, the model predicts that an investor's authentic relationships with the investment advisor and other investors strengthen the individual learning process. Consistent with Shantha, Xiaofang, and Gamini [11], these social relationships allow investors to obtain relevant information and practical knowledge to learn their mistakes. In particular, with more authentic relationships and with higher trustworthiness relationships, they feel a higher confidence regarding the information and knowledge obtained. Thus, the effect is greater in such relationships in their individual learning process. Hence, an investor's authentic relationships with the investment advisor and other investors are expected to positively moderate the relationship between the trading experience and self-reflection, as hypothesized below.



Figure 1. Conceptual framework of investor learning behavior. (Adopted from Shantha et al. [11]).

Hypothesis 4 (H4). An investor's authentic relationship with the investment advisor (ARAD) positively moderates the positive relationship between TE and SR.

Hypothesis 5 (H5). *An investor's authentic relationships with other investors (AROT) positively moderate the positive relationship between TE and SR.*

The model also assumes that an investor's desire for learning strengthens the individual learning process. The desire for learning represents the affective aspects such as interest, attention, emotions, and frustrations that influence the cognitive functioning of the brain when learning. Shantha, Xiaofang, and Gamini [11] show that these affects improve the creativity, motivation, and efficiency of learning, which strengthen the self-reflection of experiences. Accordingly, the desire for learning is expected to have a positive moderating effect on the relationship between the trading experience and self-reflection, as hypothesized below.

Hypothesis 6 (H6). An investor's desire for learning (DL) positively moderates the positive relationship between TE and SR.

Consistent with the behavioral finance and learning literature, the model argues that merely imitating others' behaviors does not produce the social form of learning. Rather it takes place when the information on strategies underlying those imitated behaviors are known to the learner [11]. Accordingly, the model predicts that the authentic relationships with other investors enable inquiry of such information in the social learning process, which means they are negatively related to herd bias occurring when trading stocks, as reflected by Hypothesis 7.

Hypothesis 7 (H7). AROT have a negative impact on HERD that occurred when trading stocks.

The investors' age, gender, and education level are incorporated in the conceptual framework as potential control variables of self-reflection. As argued by Shantha, Xiaofang, and Gamini [11] in their model of investor learning behavior, an investor's ability to be self-reflective is an outcome of his/her "mature cognitive development." Thus, these socio-demographic factors would influence the self-reflection process, as described below. The older investors are more likely to possess a higher

trading experience than younger ones, and become more self-reflective. Furthermore, men may be more interested in stock trading than women, which means, as a result, they gain a higher experience and show a greater reflective capacity. Similarly, the level of education would be positively related to the experience and self-reflection. Hence, it is important to incorporate these relationships to avoid alternative explanations and show the unique relationship between the trading experience and self-reflection. Accordingly, the model assumes that the Hypotheses 1, 4, 5, and 6 hold when controlling for the investor's age, gender, and education level.

However, the examination of the association of these socio-demographic variables with trading experience and self-reflection constructs reveals that these variables are not really confounders to be controlled in the analysis [48]. As shown in Tables 7 and 8, the age and gender are not significantly associated with the self-reflection. Even though education is associated with self-reflection, it is not related to trading experience, which is not only statistically based on the results of the one-way ANOVA test, but also in the manner it is measured in the study. Accordingly, consistent with the guidelines provided by Bernerth and Aguinis [49], these potential control variables are excluded from the analysis. Consequently, the hypotheses are examined without controlling for age, gender, and education level. The results are discussed as to whether the level of self-reflection varies with respect to these socio-demographic variables, as given by Hypothesis 8.

Hypothesis 8 (H8). There are significant differences in investors' SR level between male and female, among different age categories and education levels.

5. Research Design

5.1. Data Collection

The individual investors who are involved in trading stocks in the CSE is the unit of analysis of this study. A web-based questionnaire survey was conducted for data collection by emailing the online link of the questionnaire to investors whose security accounts have been active over the last six months. This method has advantages of having lower interviewer-bias since the participants can use their discretion when responding to the questionnaire, and is less susceptible to social desirability bias [50]. During the data collection process, conducted from March to August 2018, a total of 1000 investors randomly selected are invited to answer the questionnaire. The responses received were 189, which represents a response rate of 19%. As discussed in Section 3, the unfavorable market conditions prevailed during the period of the study could be attributed to this low response rate. The non-response bias, examined following the procedure proposed by Dooley and Lindner [51], was not found to exist in the responses received.

5.2. Characteristics of the Sample

The analysis of the demography and the investment profile of the survey participants are given in Appendix B. It shows that 71.4 percent of the respondents are male. In terms of age, about 40 percent of the respondents are below 35 years of age, while 44 percent falls in the age category of 35–54 years. Furthermore, about half of the respondents possess a bachelor's degree or higher education. The sample also includes a combination of private sector (78.3 percent) and public sector (4.8 percent) employees, retired (5.8 percent), self-employed (8.5 percent), and unemployed (2.6 percent) individuals. Hence, the sample appears to represent fairly the demographic characteristics of the individual investor population in the CSE. With respect to the respondents' trading experience which is, on average, 11 years (SD = 6.18), the sample includes 4.8 percent of investors possessing 2 years or less experience while 11.1 percent have 18 years or more of experience. Thus, it seems that data has been collected from a balanced combination of high experienced and low experienced investors. In terms of the trading frequency, 59.3 percent of the respondents trade stocks occasionally while only 9.5 percent

of them are involved in daily trading. The percentage of respondents having a low risk appetite (46.6 percent) is far greater than those with a high risk appetite (30.6 percent). As a consequence, their tendency to invest in stocks is rather low, as 20.1 percent and 48.1 percent hold, respectively, less than 5 percent and 5–15 percent of wealth in stock. Accordingly, most respondents exhibit a low risk appetite, a low trading frequency, and low stock investments, which could be attributed to the uncertain market conditions prevailed in the CSE during the study period. The uncertainty may have caused investors to become more risk-averse, which results in a shift of their investments to safer securities. This, in turn, reduces their trading frequency.

5.3. Questionnaire Design

The questionnaire consisted of a total of 14 question items, including the items related to socioeconomic characteristics such as age, gender, marital status, education and occupation, and investment profile such as trading frequency, risk appetite, and stock investment of the individual investors. When designing the questionnaire, the following procedure, as suggested by Podsakoff et al. [52], was applied to reduce the common method bias. All scales used to measure the constructs of the conceptual model were adapted from the literature and modified their wording appropriately to suit the study, as shown in Appendix A. The measurement scale of each construct was presented in a separate section of the questionnaire with different sets of instructions to reduce respondents' anxiety. The respondents were also informed that there was no right or wrong answers and their responses were kept anonymous. In addition, the questionnaire was pre-tested before the survey in a sample of 30 individual investors in an attempt to enhance its face and content validity. Furthermore, the meaning and wording of the measurement items were discussed with three academics and three investment advisors to further ensure the clarity of the questions asked and instructions provided to answer the questionnaire. Moreover, Harman's one-factor test reveals that the common method bias is absent in data.

5.4. Measurement Scales

Appendix A provides an overview of the measurement scales used to measure the constructs of the conceptual model. Consistent with the studies of Abreu and Mendes [53] and Mishra and Metilda [54], the trading experience was measured by asking the participants to mention the number of years over which they had been trading stocks on the CSE. The work of Kember et al. [55] was used to measure the extent to which the investors exercise their self-reflection for learning. Of the three levels of reflection manifested in their scale, this study adapted the items relating to the process and premise reflection levels. The process reflection refers to how a person perceives, thinks, feels, and acts in response to his/her past experiences, whereas the premise reflection, the deepest level of reflection, results in identifying psychological and cultural limitations in one's established frame of reference [56]. Accordingly, the self-reflection was measured by seven items, consisting of three items associated with the process reflection and four items related to the premise reflection. The extent of the survey participants inclined to herd bias was assessed following the studies of Waweru et al. [57] and Kengatharan and Kengatharan [58]. Waweru, Munyoki, and Uliana [57] found that investors tend to trade around stock's price changes, popularity, and past trends, which result in herd behavior among investors. Kengatharan and Kengatharan [58] used these behavioral factors in their scale to measure herd bias of the individual investors of the CSE. Consistent with these works, three items, representing these three behavioral characteristics, were used to measure herd bias of the respondents.

Relating to the moderating variables, the following measurement scales were adapted. The assessment of the participants' desire for learning was based on the self-directed learning readiness scale proposed by Fisher et al. [59]. Their scale originally included 12 items to measure the desire for learning. However, the findings of Fisher and King [60] and Williams and Brown [61] confirmed for a 10-item scale that showed a better model fit when compared to the original 12-item scale. Of these 10 items, this study adapted only eight items since two items were dropped due to low factor loading,

as indicated by loading relevant test [62]. Authentic relationships with the investment advisor and other investors were each measured by a five-item scale based on the work of Kale et al. [63]. However, one item was removed when measuring an authentic relationship with the investment advisor due to low factor loading.

5.5. Methodology

The conceptual model, discussed in Section 4, intends to predict the determinants of investors' learning behavior and explain their effects on herd bias. The literature, for example, Becker et al. [64], Evermann and Tate [65], and Sarstedt et al. [66] suggest to use the Partial Least Squares Structural Equation Modeling (PLS-SEM) when the goal of the analysis is to predict and explain a target construct and identify its predecessors. Accordingly, the data analysis was conducted using the PLS-SEM, supported by SmartPLS 3 software. When compared with the factor-based SEM, the PLS-SEM has advantages such as higher statistical power, which is appropriate in the case of exploratory research, ability to handle complex models with many constructs and indicator variables, consistent estimates of parameters, and flexibility in terms of sample size and parametric distributional assumptions such as multivariate normality (Sarstedt, Ringle and Hair [66], Hair, et al. [67], and references therein).

Consistent with the procedure suggested by Sarstedt et al. [68], the following two steps were taken in the data analysis process. First, the measurement model was evaluated to ensure the measurement quality of the model's constructs. After that, the structural model was evaluated for its predictive capability and testing the hypothesized relationships. Since the constructs were reflectively specified, their measurement quality was evaluated in terms of indicator reliability, internal consistency reliability, convergent validity, and discriminant validity. In the second step, when evaluating the structural model, the model was checked for collinearity issues by analyzing the variance inflation factor (VIF). Afterward, the model's predictive capability, as reflected by the coefficient of determination (R^2), cross-validated redundancy (Q^2), and effect-size (f^2), was assessed and the hypotheses were examined to infer the determinants of the learning behavior and their effects on herd bias. The model's predictive accuracy, as indicated by Q^2 , was estimated based on the blindfolding procedure with an omission distance of six [69–71]. f^2 , a measure of the effect-size of a particular predictor variable on its endogenous variable, was calculated by following the procedure suggested by Henseler and Chin [72]. The significance of path coefficients for hypothesis testing was estimated using 5000 bootstrap subsamples at the significance level of 5 percent [67].

6. Reliability and Validity of Measurements

The results related to the assessment of the reliability and validity of the constructs are provided in Tables 1–3, and Appendix C. After performing the loading relevant test [62], indicator items demonstrate a satisfactory level of their reliability since, except for a few indicators, all other indicators show loading values greater than 0.7 on their respective constructs, as given in Table 1 [73]. In addition, Cronbach's Alpha and Composite Reliability values of all the constructs exceed the 0.7 level, which indicates their internal consistency reliability [74,75]. Furthermore, all the constructs exhibit an average variance extracted value (AVE) in excess of the cut-off level of 0.5 for their convergent validity. The discriminant validity was assessed by referring to Fornell and Larcker criterion, cross loadings, and Heterotrait-monotrait (HTMT) criterion, as follows. Table 2 shows that the square root of AVE values of all the constructs are higher than their correlation coefficients with other constructs [76]. As given in Appendix C, each indicator loads at the highest value to the construct to which it relates [77,78]. The HTMT ratio of correlations, provided in Table 3, are less than the 0.85 level [79]. Hence, these results provide strong evidence of discriminant validity of the model's constructs. Moreover, the VIF values, which are shown in Table 4, do not indicate a multi-collinearity issue in the model since they are less than five [80,81].

Construct	Indicator Item	Indicator Loading	Cronbach's Alpha	CR	AVE
ARAD	Arad_1	0.866	0.875	0.891	0.671
	Arad_2	0.777			
	Arad_3	0.811			
	Arad_5	0.820			
AROT	Arot_1	0.628	0.851	0.890	0.622
	Arot_2	0.842			
	Arot_3	0.838			
	Arot_4	0.819			
	Arot_5	0.794			
DL	Dl_1	0.799	0.912	0.928	0.618
	Dl_2	0.819			
	Dl_3	0.806			
	Dl_4	0.827			
	Dl_6	0.748			
	Dl_7	0.733			
	Dl_8	0.763			
	Dl_9	0.788			
HERD	Herd_1	0.875	0.836	0.891	0.732
	Herd_2	0.861			
	Herd_3	0.830			
SR	Sr_1	0.567	0.879	0.883	0.527
	Sr_2	0.535			
	Sr_3	0.819			
	Sr_4	0.838			
	Sr_5	0.649			
	Sr_6	0.815			
	Sr_7	0.789			
TE	TradeYrs	1.000	1.000	1.000	1.000

Table 1. Assessment of the measurement quality of the model's constructs.

Note: This table presents indicator items' loading, cronbach's alpha, composite reliability (CR), and average variance extracted (AVE) values for assessing the measurement quality of the model's constructs. An indicator loading value greater than 0.5 shows the indicator reliability [73]. A set of indicators to measure each construct is found from the loading relevant test [62]. The Cronbach's alpha and CR values greater than 0.7 indicate the internal consistency reliability [74,75]. The AVE value greater than 0.5 represents the convergent validity [76,82].

	ARAD	AROT	DL	HERD	SR	TE	Discriminant Validity Met?
ARAD	0.819						Yes
AROT	0.415	0.788					Yes
DL	0.421	0.523	0.786				Yes
HERD	-0.099	0.145	-0.173	0.855			Yes
SR	0.336	0.294	0.542	-0.310	0.726		Yes
TE	0.101	0.157	0.185	-0.011	0.205	Single item	Yes

Note: This table reports the square root of average variance extracted (AVE) value for each construct and its correlations with other constructs. The square root of AVE values is shown on the diagonal and printed in bold. The non-diagonal elements represent correlations of a construct with other constructs. The discriminant validity is met when the square root of AVE of a construct is greater than its correlation coefficients with other constructs [76].

	ARAD	AROT	DL	HERD	SR	TE
ARAD						
AROT	0.463					
DL	0.456	0.589				
HERD	0.165	0.186	0.212			
SR	0.353	0.324	0.597	0.353		
TE	0.117	0.172	0.193	0.014	0.226	

Table 3. HTMT criterion analysis for assessing discriminant validity.

Note: The HTMT ratio of correlations between the model's constructs are reported in this table. The HTMT value of less than 0.85 indicates discriminant validity [79].

Table 4. VIF values for examining multi-collinearity.

	ARAD	AROT	DL	SR	TE
SR	1.375	1.504	1.692		1.116
HERD		1.094		1.094	

Note: This table shows the variance inflation factor (VIF) values of exogenous constructs (given in the column) with respect to their endogenous constructs (given in raw wise) for the examination of multicollinearity. The VIF value of less than 5 indicates the absence of multi-collinearity [80,81].

7. Discussion of Results on Learning Behavior

Figure 2 depicts the main results relating to the investors' learning behavior examined in this paper. R^2 values of SR and HERD constructs are, respectively, 37.5 percent and 14.8 percent. Q^2 values of SR and HERD constructs are respectively 0.176 and 0.097, which indicate an acceptable level of the path model's predictive accuracy and relevance [66]. The estimates of path coefficients, their significance, and effect sizes are discussed below to test hypotheses related to individual and social learning behaviors.



Figure 2. Main results of the investors' learning behavior. Note: The significance at 1 percent and 5 percent levels are denoted by ** and *, respectively.

7.1. Investors' Individual Learning Behavior

Concerning the results related to the individual learning behavior, presented in Table 5, the trading experience has a significant positive impact on the extent of self-reflection, which is consistent with H1. An increase in one standard deviation of the TE construct increases the SR construct by an 18.9 percent standard deviation ($f^2 = 0.048$, p < 0.01). In addition, confirming H2, the results reveal that an increase in one standard deviation of the SR construct decreases the HERD construct by a 38.9 percent standard deviation ($f^2 = 0.161$, p < 0.01). In line with H3, the SR construct mediates the relationship between TE and HERD constructs at a 5 percent level of significance. However, as shown in part A of Table 5, the evidence does not support a direct effect of the trading experience on minimizing

herd bias (TE \rightarrow HERD).	Thus, it confirms the	e full-mediation of SR	on the relationship	between TE	and
HERD [83].					

Hypothesis	Path	Path Coefficient	Standard Error	t-Value	<i>p</i> -Value	Decision	f^2
Part A: Effec	t of the trading exper	ience on self-re	eflection and	herd bias			
H1	$TE \rightarrow SR$	0.189	0.068	2.640	0.004 **	Accept	0.048
H2	$SR \rightarrow HERD$	-0.389	0.090	4.294	0.000 **	Accept	0.161
H3	$\text{TE}{\rightarrow}\text{SR}{\rightarrow}\text{HERD}$	-0.069	0.033	2.080	0.019 *	Accept	
	$\text{TE}{\rightarrow}\text{HERD}$	-0.017	0.117	0.153	0.439	-	
Part B: Mode	erating effect of an au	thentic relatior	nship with the	e investment	advisor on	self-reflectic	n
H4	$ARAD \times TE \rightarrow SR$	-0.085	0.085	1.012	0.156	Reject	0.006
	$\begin{array}{c} \text{ARAD} \times \text{TE} \rightarrow \\ \text{SR} \rightarrow \text{HERD} \end{array}$	0.033	0.034	0.977	0.164	neject	
	$ARAD \rightarrow SR$	0.068	0.067	1.017	0.154		0.006
	$\begin{array}{c} \text{ARAD} \rightarrow \text{SR} \rightarrow \\ \text{HERD} \end{array}$	-0.026	0.024	1.025	0.153		
Part C: Mode	erating effect of the a	uthentic relatio	onship with o	ther investor	s on self-refl	ection	
H5	$\text{AROT} \times \text{TE} {\rightarrow} \text{SR}$	-0.175	0.083	2.169	0.015 *	Reject	0.027
	$\begin{array}{l} \text{AROT} \times \text{TE} \rightarrow \\ \text{SR} \rightarrow \text{HERD} \end{array}$	0.069	0.037	1.856	0.032 *	10,000	
	$\text{AROT} \rightarrow \text{SR}$	-0.028	0.073	0.432	0.333		0.001
	$\begin{array}{c} \text{AROT} \rightarrow \text{SR} \rightarrow \\ \text{HERD} \end{array}$	0.011	0.028	0.454	0.325		
Part D: Mod	erating effect of desir	e for learning o	on self-reflect	ion			
H6	$DL \times TE \rightarrow SR$	-0.094	0.100	0.931	0.176	Reject	0.007
	$DL \times TE \rightarrow SR \rightarrow HERD$	0.038	0.042	0.865	0.194	Ingeet	
	$DL \rightarrow SR$	0.404	0.082	5.016	0.000 **		0.165
	$\text{DL}{\rightarrow}\text{SR}{\rightarrow}\text{HERD}$	-0.159	0.053	2.986	0.001 **		

Table 5. Estimates of the model's path coefficients, their significance, and effect sizes relating to individual learning behavior.

Note: This table reports the hypothesis testing results related to individual learning behavior. The model hypothesizes that TE is positively related to SR (H1), which, in turn, is negatively related to HERD (H2). It also assumes SR mediates the relationship between TE and HERD (H3). Furthermore, ARAD, AROT, and DL have moderating effects on the relationship between TE and SR (as reflected by H4, H5, and H6, respectively). The significance at 1 percent and 5 percent levels are denoted by ** and *, respectively. f^2 denotes the effect-size of the path's exogenous variable on its endogenous variable. As a rule of thumb, f^2 values of 0.02, 0.15, and 0.35 represent the cut-off values for small, medium, and large effects [84].

Then, moderating effects of ARAD, AROT, and DL constructs on the relationship between TE and SR, as reflected by H4, H5, and H6, are assessed. The estimates, given in parts B, C, and D of Table 5, reveal that these moderating effects are absent in the individual learning process during the period of the study. However, as shown in part D, the DL construct has a direct positive effect on the SR construct ($f^2 = 0.165$, p < 0.01), which, in turn, has a negative impact on the HERD construct (p < 0.01). Hence, rather than being a moderating variable, DL should be considered as a direct predictor of SR since it has a negative effect on herd bias through the mediating effect of self-reflection.

Next, the size of the effects of predictor constructs on respective endogenous constructs, as reflected by f^2 given in Table 5, are examined. Accordingly, the DL construct has the largest effect on the SR construct with an f^2 value of 0.165, which can be classified as a medium effect based on the cut-off values given by Cohen [84]. Nevertheless, the effect size of TE on SR appears to be small ($f^2 = 0.048$), while the moderating variables—ARAD and AROT—indicate no effects. The small effect size of TE on SR could be attributed to the uncertain market conditions prevailed during the period of the study. As discussed in Section 5.2, the uncertainty caused investors to become more risk averse

and reduce their stock holding, which resulted in a lower trading frequency during this period. As a consequence, investors exhibited a lower tendency to be involved in stock trading, which shows a small effect of their trading experience in the self-reflection process. In addition, the lower trading frequency would have reduced their interaction with the investment advisor to a considerable level, which could be attributed to the absence of the moderating effect of ARAD in the self-reflection of experiences. Furthermore, as a frontier market, unsophisticated investors dominate the CSE. Hence, an investor's peer-relationships (AROT) may not provide quality information to strengthen his/her learning process, which results in an absence of the moderating effect of AROT in the self-reflection process.

7.2. Investors' Social Learning Behavior

The model assumes that the AROT construct is negatively related to the HERD construct in an investor's social learning process, as reflected by hypothesis H7. The results, reported in Table 6, do not provide support for this hypothesis, which indicates the absence of social learning behavior among the respondents. It is, however, interesting to note that the results rather confirm for an increase of herd bias from the peer-relationships, as reflected by the positive significant coefficient of the path AROT \rightarrow HERD. Accordingly, it shows that an increase in one standard deviation of the AROT construct increases the HERD construct by a 27.3 percent standard deviation ($f^2 = 0.072$, p < 0.01). The uncertain market conditions prevailed during the period of the study and the dominance of unsophisticated investors could be considered as possible explanations for this increased herd tendency from peer-relationships. Therefore, in line with the herding literature, it is evident that uncertain market conditions motivate investors to disregard their own information and imitate others' behaviors when trading stocks.

Table 6. Estimates of the model's path coefficient, its significance, and effect size relating to social learning behavior.

Hypothesis	Path	Path Coefficient	Standard Error	<i>t</i> -Value	<i>p</i> -Value	Decision	f^2
H7	$\text{AROT}{\rightarrow}\text{HERD}$	0.273	0.077	3.339	0.000 **	Reject	0.072
					1 1		

Note: This table reports the hypothesis testing results related to the social learning behavior. The model hypothesizes that AROT is negatively related to HERD (as reflected by H7). The significance at 1 percent is indicated by **. f^2 denotes the effect-size of the path's exogenous variable on its endogenous variable. As a rule of thumb, f^2 values of 0.02, 0.15, and 0.35 represent the cut-off values for small, medium, and large effects [84].

7.3. Diversity of Learning with Respect to Investors' Demography

The hypothesis H8 is concerned with finding whether the level of SR varies between male and female respondents, and among their different age groups and education levels. The Independent sample *t*-test was conducted to check whether SR is significantly different between male and female investors (Table 7), while the one-way ANOVA test was performed to find whether SR significantly varies among different age groups and education levels (Table 8). The results show that the level of SR varies among different education levels. However, the same is not evident between male and female respondents and among their different age categories.

Table 7. Test of equality of SR between male and female investors.

Investor Group	Mean	Standard Deviation	Standard Error of Mean	<i>t</i> -Value	<i>p</i> -Value
Male Female	3.926 3.770	0.559 0.802	0.048 0.109	1.521	0.130

Note: This table shows the results of the independent sample t-test conducted for examining whether SR is different between male and female investors. The null hypothesis that mean values between male and female groups are equal, is not rejected at the 5 percent level of significance.

		Sum of Squares	Mean Square	F-Value	<i>p</i> -Value
Age and SR	Between Groups	0.884	0.221	0.535	0.710
0	Within Groups	76.026	0.413		
Education and	Between Groups	17.359	4.340	13.409	0.000 **
SR	Within Groups	59.551	0.324		

Table 8. Test of equality of SR among different age groups and education levels.

Note: This table presents the results of the one-way ANOVA test to examine whether SR varies among different age groups and education levels. The significance at 1 percent level is denoted by **.

7.4. Comparison of Results with Previous Studies on the CSE

The results of this study show that investors' peer relationships drive herd behavior, whereas their self-reflection of experiences reduces herding when trading stocks. When concerning the magnitude of these two contradictory effects on herding, the effect-size of the self-reflection ($f^2 = 0.161$) appears to be higher than that of the peer relationships ($f^2 = 0.072$), which indicates a declining herd tendency at the aggregate market level. These results support the findings of the studies conducted by Shantha [8] and Xiaofang and Shantha [10]. Both of these studies find that herd tendency is strong during the periods of political uncertainty of the country (2000–2009) and the market bubble and crash (2009–2012), whereas it declines and disappear afterwards. Therefore, the findings of this study confirm that the investors' learning from their past experiences is the main reason for herding to decline and disappear during the past few market periods. Accordingly, it is evident that investors have learned the irrationality of herding from their past experiences. As a consequence, they tend to shift away from such irrational behavior when trading stocks.

8. Conclusions and Implications

To the best of my knowledge, this work is the first to show the link between the learning behavior of individual investors and its impact on herd bias. In view of that, it extends the previous studies of Shantha [8] and Xiaofang and Shantha [10] by providing empirical evidence on factors that ground for the declining herd behavior in the market. Of the eight hypotheses formulated in this study, six hypotheses (from H1 to H6) intend to examine the individual learning behavior as well as one hypothesis (H7) for the social learning behavior and one hypothesis (H8) for the analysis of the learning-diversity with respect to the demography of the investors. The main conclusion are as follows.

- The results confirm the hypotheses from H1 to H3, indicating a full mediation effect of investors' self-reflection on the relationship between their trading experience and herd bias. Hence, challenging the reinforcement learning (trial-and-error behavior) assumed by the previous studies conducted on the agent-based financial markets, the findings reveal that past trading experiences do not directly produce learning. Rather, the experiences are to be cognitively reflected (that is, self-reflection) to yield learning to reduce behavioral biases.
- However, findings do not support the moderating effects of investors' authentic relationships with
 investment advisors and peer investors, as reflected by the hypotheses H4 and H5, respectively.
 The uncertainty of market conditions caused the investors to reduce their risk appetite, which
 led to a low level of stock holding and trading frequency prevailing during this period. As a
 consequence, they may have maintained a low level of interactions with their investment
 advisors, which results in an absence of learning effects from the relationships with the advisors.
 Furthermore, since the CSE is a frontier stock market, a category of markets typically dominated
 by unsophisticated investors, the learning effect is absent through their peer-relationships.
 Accordingly, this evidence indicates that market conditions affect the extent of learning that
 occurred within an individual investor.

- Despite the absence of the moderating effect, as reflected by the hypothesis H6, the evidence shows that an investor's desire for learning has a direct influence on the self-reflection process. Accordingly, during the period of the study, the individual learning appears to have taken place through the self-reflection of past trading experiences, which is induced by the desire for learning. Furthermore, supporting the hypothesis H8, the results reveal that the extent of the self-reflection varies with respect to the investor's level of education.
- Invalidating the hypothesis H7, the findings indicate that the social learning is absent among the investors due to the dominance of unsophisticated investors in the market. It is evident that herd bias tends to increase among investors through these peer-relationships.

The findings of this paper also provide the following implications for practice.

- Consistent with the AMH, the success of an investor is highly dependent on the ability to learn and adapt to dynamic market conditions with feasible investment strategies. Stock exchanges conduct regularly educational programs to improve the financial literacy of investors to facilitate them to achieve a higher investment performance. The findings suggest that these educational initiatives should be designed to empower them to learn by reflecting on their own experiences. Consequently, they will be able to effectively learn from their past trading experiences, which reduced the exposure to behavioral biases when trading stocks. Accordingly, enhancing the self-reflection capacity of investors should be a key focus of these educational programs. The increased sophistication of investors and their stock market participation will take place while enabling them to engage in social learning through their peer-relationships.
- The investment advisors should continuously involve in building up strong client-relationships by strengthening the interaction, cooperation, and mutual trust with their clients. As a result, the clients will regularly interact with their advisors irrespective of the market conditions. It will support investors to improve their self-reflection capacity and arrive at better investment strategies for adapting to dynamic market conditions.

Hence, when the above implications are incorporated into the initiatives of stock exchanges and investment advisors, it can be expected that the sophistication of investors and their stock market participation will be enhanced. As stated in Sections 1 and 2.1, these will improve the ability of investors to recognize companies' sustainable development endeavor, which facilitates the companies to finance such developments at a lower cost. Accordingly, the investor learning behavior would eventually result in broader access to finance at lower cost so that sustainable economic development could be promoted.

9. Limitations and Directions for Future Research

This study has been conducted on a frontier market focusing a period over which its trading environment is highly uncertain. Hence, in terms of the type of the market and its environmental conditions, it is an ideal context for studying herd bias. However, it is subject to the following limitations. The findings of this study may not be generalizable to developed and emerging markets due to differences in investment and regulatory environment operating in those markets. In addition, the learning behavior has been studied by integrating its effects on herd bias, which is one of many forms of behavioral biases to which investors are exposed when trading stocks. Thus, the nature and extent of the learning may be different when the other types of behavioral biases are examined. Furthermore, the unit of analysis considered in this study is the individual investor. Hence, the results of this study may not be generalizable to predict the learning behavior of other investor-types such as institutional investors. Accordingly, future works can contribute to overcome these limitations by extending similar studies to other categories of markets (developed and emerging markets), different forms of behavioral biases (for example, heuristic and prospect biases), and other investor-types (for example, institutional investors and financial analysts). Funding: This research received no external funding.

Conflicts of Interest: The author declares no conflict of interest.

Appendix A. Measurement Scales

Construct and Item Code	Item Wording	Source
	Trading experience (TE)	
TradeYrs	How long have you been investing in the stock market? (State in number of years)	Abreu and Mendes [53], Mishra and Metilda [54]
	Self-reflection (SR)	
	How would you respond to your past stock trading experiences?	
Sr_1	I sometimes question the way others do trading and try to think of a better way.	Kember, Leung, Jones,
Sr_2	I like to think over what I have been doing and consider alternative ways of doing it.	Loke, McKay, Sinclair, Tse, Webb, Yuet Wong and
Sr_3	I often evaluate my past stock trading so I can learn from it and improve my next trading experience.	Wong [55]
Sr_4	As a result of my trading experience, I have changed the way I make trading decisions.	
Sr_5	My experience has challenged some of my firmly held ideas and beliefs.	
Sr_6	As a result of the experience, I have changed the way I trade stock.	
Sr_7	I have discovered faults in what I had previously believed to be right. (1 = Strongly disagree, 5 = Strongly agree)	
	Herd bias (HERD)	
	Please indicate the extent to which you agree with the following. I would invest stock by following my friends'	
Herd_1	recommendations.	Waweru, Munyoki and
Herd_2 Herd_3	I would buy the stocks whose prices have risen for a period. I would follow the market trend when buying/selling	Uliana [57]
	stocks. (1 = Strongly disagree, 5 = Strongly agree)	
	Desire for learning (DL)	
Dl_1 Dl 2	Please indicate to what extent you feel about the following. I want to learn new information I enjoy learning new information	Fisher and King [60]
	I have a need to learn	
Dl_4	I enjoy a challenge	
DI_5	I do not enjoy studying	
DI_6	Lloarn from my mistakes	
D1_7 D1_8	I need to know why	
D1 9	Lam open to new ideas	
Dl_10	When presented with a problem I cannot resolve, I will ask for assistance (R) (1 = Strongly disagree, 5 = Strongly agree)	

Table A1. Question items of the model's constructs.

Construct and Item Code	Item Wording	Source
Autho	entic relationship with investment advisor (ARAD)	
	How would you describe your relationship with your investment advisor?	
Arad 1	I would let my adviser decide everything.	
Arad_2	I prefer to ask my adviser's opinion for trading.	Kale, Singh and Perlmutter
Arad_3	I would trust my adviser.	[63]
Arad_4	My adviser provides me with information important to make my trading decisions.	
Arad_5	My adviser cooperates and shares ideas, feelings, beliefs, etc. (1 = Never, 2 = Rarely, 3 = Sometimes, 4 = Very often, 5 = Always)	
Au	thentic relationship with other investors (AROT)	
	How would you describe your relationships with other investors?	
Arot 1	Friendly and can talk about difficulties personally	
Arot 2	Mutually trusting	Kale, Singh and Perlmutter
Arot_3	Mutually respectful	[63]
Arot_4	Highly give-and-take	
Arot_5	Share ideas, feelings, beliefs, etc.	
	(1 = Never, 2 = Rarely, 3 = Sometimes, 4 = Very often, 5 =	
	Always)	
	Socio-demography	
Age	Please indicate your age (less than 25 years, 25–34 years, 35–44 years, 45–54 years, and 55 years or above)	
Gender	Please indicate your gender (male, female)	
Marital status	Please indicate your marital status (married, unmarried)	
	Please indicate your highest academic qualification (O/L,	
Education	A/L, diploma, degree, postgraduate diploma, MBA/MSc, PhD)	
	Please state your highest professional qualification, if any.	
	Please indicate your current occupation (Private sector	
Occupation	employee, public sector employee, retired, self-employed, unemployed)	
	Investment profile	
Trading	How often do you buy or sell stocks? (occasionally, once a	
frequency	month, once a week, 2–3 times a week, daily)	
	How do you think your best friend would describe you?	
	unwilling to take risks	
	willing to take modest risks but only after careful	
	consideration and professional advisement	
Risk appetite	willing to take modest risks after some thought	
	willing to take substantial risks after careful consideration	
	and professional advisement	
	someone who embraces risk, perhaps without sufficient	
	consideration	
Stock-holding	Please indicate the percentage of your wealth invested in stocks?	

Table A1. Cont.

		No. of Participants	Percentage of Respondents	
Gender	Male	135	71.4%	
	Female	54	28.6%	
Age	Less than 25 years	13	6.9%	
0	25–34 years	64	33.9%	
	35–44 years	46	24.3%	
	45–54 years	38	20.1%	
	55 years or above	28	14.8%	
Marital Status	Married	131	69.3%	
	Unmarried	58	30.7%	
Highest academic qualification	A/L	44	23.3%	
	Diploma 46		24.3%	
	Degree	59	31.2%	
	Postgraduate Diploma	10	5.3%	
	MBA/MSc	30	15.9%	
	Ph.D	0	0.0%	
Occupation	Private sector employee	148	78.3%	
	Public sector employee	9	4.8%	
	Retired	11	5.8%	
	Self-employed	16	8.5%	
	Unemployed	5	2.6%	

 Table A2. Demography of Survey Respondents.

Appendix B. Demography and Investment Profile of Survey Respondents

 Table A3. Respondents' Investment Profile.

		No. of Participants	Percentage of Respondents	
Trading experience	2 years or less	9	4.8%	
	3–7 years	46	24.3%	
	8–12 years	79	41.8%	
	13–17 years	34	18.0%	
	18 years or above	21	11.1%	
Trading frequency	Occasionally	112	59.3%	
	Once a month	17	9.0%	
	Once a week	18	9.5%	
	2–3 times a week	24	12.7%	
	Daily	18	9.5%	
Risk Appetite	Very low risk taker	26	13.8%	
	Low risk taker	62	32.8%	
	Average risk taker	43	22.8%	
	High risk taker	53	28.0%	
	Very high risk taker	5	2.6%	
Duen entire of succelth	Less than 5%	38	20.1%	
Proportion of wealth	5-15%	91	48.1%	
invested in stocks	16-25%	26	13.8%	
	26-40%	11	5.8%	
	41-60%	15	8.0%	
	More than 60%	8	4.2%	

Appendix C. Cross-Loadings of Indicator Items to Latent Variables

The following table presents the loading values of each indicator item to its own construct and other constructs of the model. The discriminant validity is established when the loadings of indicator items to its own construct are higher than those of other constructs [77,78].

Construct	Indicator Item	ARAD	AROT	DL	HERD	SR	TE
ARAD	Arad_1	0.866	0.378	0.437	-0.121	0.314	0.043
	Arad_2	0.777	0.188	0.225	-0.124	0.161	0.096
	Arad_3	0.811	0.307	0.303	-0.025	0.230	0.135
	Arad_5	0.820	0.410	0.353	-0.076	0.332	0.081
AROT	Arot_1	0.344	0.628	0.312	-0.028	0.267	0.125
	Arot_2	0.276	0.842	0.344	0.263	0.131	0.096
	Arot_3	0.373	0.838	0.525	0.096	0.272	0.114
	Arot_4	0.277	0.819	0.418	0.170	0.217	0.072
	Arot_5	0.347	0.794	0.430	0.047	0.263	0.213
DL	Dl_1	0.319	0.316	0.799	-0.198	0.456	0.112
	Dl_2	0.354	0.461	0.819	-0.122	0.433	0.169
	Dl_3	0.369	0.378	0.806	-0.121	0.404	0.175
	Dl_4	0.288	0.434	0.827	-0.078	0.340	0.100
	Dl_6	0.292	0.467	0.748	-0.125	0.472	0.167
	Dl_7	0.365	0.386	0.733	-0.164	0.434	0.181
	Dl_8	0.279	0.389	0.763	-0.134	0.431	0.079
	Dl_9	0.363	0.452	0.788	-0.162	0.403	0.174
HERD	Herd_1	-0.111	0.069	-0.127	0.875	-0.259	-0.017
	Herd_2	0.015	0.232	-0.055	0.861	-0.250	0.001
	Herd_3	-0.193	0.035	-0.289	0.830	-0.290	-0.015
SR	Sr_1	0.112	0.066	0.341	-0.132	0.567	0.034
	Sr_2	0.121	0.119	0.318	-0.089	0.535	0.224
	Sr_3	0.293	0.279	0.508	-0.285	0.819	0.190
	Sr_4	0.333	0.260	0.445	-0.384	0.838	0.122
	Sr_5	0.140	0.110	0.223	-0.198	0.649	0.143
	Sr_6	0.308	0.199	0.377	-0.189	0.815	0.185
	Sr_7	0.283	0.350	0.460	-0.207	0.789	0.153
TE	TradeYrs	0.101	0.157	0.185	-0.011	0.205	1.000

Table A4. Cross-loadings of indicator items.

References

- 1. Brundtland, G.H. *Our Common Future: Report of the 1987 World Commission on Environment and Development;* Oxford University Press: Oxford, UK, 1987.
- 2. Waygood, S. How do the capital markets undermine sustainable development? What can be done to correct this? *J. Sustain. Financ. Investig.* **2011**, *1*, 81–87. [CrossRef]
- 3. Lo, A.W. The adaptive markets hypothesis: Market efficiency from an evolutionary perspective. *J. Portf. Manag.* **2004**, *30*, 15–29. [CrossRef]
- 4. Lo, A.W. Reconciling efficient markets with behavioral finance: The adaptive markets hypothesis. *J. Invest. Consult.* **2005**, *7*, 21–44.
- 5. Lo, A.W. Adaptive Markets and the New World Order. Financ. Anal. J. 2012, 68, 18–29. [CrossRef]
- 6. Kumar, S.; Goyal, N. Behavioural biases in investment decision making—A systematic literature review. *Qual. Res. Financ. Mark.* **2015**, *7*, 88–108. [CrossRef]
- Yao, J.; Ma, C.; He, W.P. Investor herding behaviour of Chinese stock market. *Int. Rev. Econ. Financ.* 2014, 29, 12–29. [CrossRef]
- 8. Shantha, K.V.A. Shifts in Herd Mentality of Investors in Uncertain Market Conditions: New Evidence in the Context of a Frontier Stock Market. *J. Econ. Behav. Stud.* **2018**, *10*, 203–219. [CrossRef]

- 9. Guney, Y.; Kallinterakis, V.; Komba, G. Herding in frontier markets: Evidence from African stock exchanges. *J. Int. Financ. Mark. Inst. Money* **2017**, *47*, 152–175. [CrossRef]
- 10. Xiaofang, C.; Shantha, K.V.A. Are market performance and volatility determining the evolution of herd mentality of investors in a frontier stock market? Evidence from the Colombo Stock Exchange of Sri Lanka. *J. Manag. Matters* **2018**, *5*, 1–12.
- Shantha, K.V.A.; Xiaofang, C.; Gamini, L.P.S. A Conceptual Framework on Individual Investors' Learning Behavior in the Context of Stock Trading: An Integrated Perspective. *Cogent Econ. Financ.* 2018, 6, 1–22. [CrossRef]
- 12. Economou, F. Herd behavior in frontier markets: Evidence from Nigeria and Morocco. In *Handbook of Frontier Markets;* Elsevier: Amsterdam, The Netherlands, 2016; pp. 55–69.
- 13. Bikhchandani, S.; Sharma, S. Herd behavior in financial markets. IMF Staff Pap. 2000, 47, 279–310.
- 14. Hirshleifer, D.; Hong, T.S. Herd behaviour and cascading in capital markets: A review and synthesis. *Eur. Financ. Manag.* **2003**, *9*, 25–66. [CrossRef]
- 15. Spyrou, S. Herding in financial markets: A review of the literature. *Rev. Behav. Financ.* **2013**, *5*, 175–194. [CrossRef]
- 16. Balcilar, M.; Demirer, R.; Hammoudeh, S. What drives herding in oil-rich, developing stock markets? Relative roles of own volatility and global factors. *N. Am. J. Econ. Financ.* **2014**, *29*, 418–440. [CrossRef]
- 17. Ito, M.; Noda, A.; Wada, T. The evolution of stock market efficiency in the US: A non-Bayesian time-varying model approach. *Appl. Econ.* **2016**, *48*, 621–635. [CrossRef]
- 18. Choe, H.; Kho, B.-C.; Stulz, R.M. Do foreign investors destabilize stock markets? The Korean experience in 1997. *J. Financ. Econ.* **1999**, *54*, 227–264. [CrossRef]
- 19. Hwang, S.; Salmon, M. Market stress and herding. J. Empir. Financ. 2004, 11, 585–616. [CrossRef]
- 20. Nguyen, T.T. *Herding Behaviour in Vietnamese Stock Market;* Department of Finance and Statistics, Hanken School of Economics: Helsinki, Finland, 2018.
- 21. Chan, N.T.; LeBaron, B.; Lo, A.W.; Poggio, T. Agent-based models of financial markets: A comparison with experimental markets. In *Working Paper MIT Artificial Markets Project*; MIT: Cambridge, MA, USA, 1999.
- 22. Yamamoto, R. Evolution with individual and social learning in an agent-based stock market. In Proceedings of the 11th International Conference on Computing in Economics and Finance, Washington, DC, USA, 11–13 July 2003.
- 23. Hommes, C.H. Heterogeneous agent models in economics and finance. In *Handbook of Computational Economics*, 1st ed.; Tesfatsion, L., Judd, K.L., Eds.; Elsevier: Amsterdam, The Netherlands, 2006; Volume 2, pp. 1109–1186.
- 24. LeBaron, B. Agent-based computational finance. In *Handbook of Computational Economics*; Tesfatsion, L., Judd, K.L., Eds.; Elsevier: Amsterdam, The Netherlands, 2006; Volume 2, pp. 1187–1233.
- 25. LeBaron, B. Active and passive learning in agent-based financial markets. *East. Econ. J.* **2011**, *37*, 35–43. [CrossRef]
- 26. LeBaron, B. Heterogeneous gain learning and the dynamics of asset prices. *J. Econ. Behav. Organ.* **2012**, *83*, 424–445. [CrossRef]
- 27. Bossan, B.; Jann, O.; Hammerstein, P. The evolution of social learning and its economic consequences. *J. Econ. Behav. Organ.* **2015**, *112*, 266–288. [CrossRef]
- Pastore, A.; Esposito, U.; Vasilaki, E. Modelling stock-market investors as reinforcement learning agents. In Proceedings of the 2015 IEEE International Conference on Evolving and Adaptive Intelligent Systems, Douai, France, 1–3 December 2015; pp. 1–6.
- 29. Gervais, S.; Odean, T. Learning to be overconfident. Rev. Financ. Stud. 2001, 14, 1–27. [CrossRef]
- 30. Feng, L.; Seasholes, M.S. Do investor sophistication and trading experience eliminate behavioral biases in financial markets? *Rev. Financ.* 2005, *9*, 305–351. [CrossRef]
- 31. Dhar, R.; Zhu, N. Up close and personal: Investor sophistication and the disposition effect. *Manag. Sci.* **2006**, 52, 726–740. [CrossRef]
- 32. Seru, A.; Shumway, T.; Stoffman, N. Learning by trading. Rev. Financ. Stud. 2009, 23, 705–739. [CrossRef]
- Nicolosi, G.; Peng, L.; Zhu, N. Do individual investors learn from their trading experience? *J. Financ. Mark.* 2009, 12, 317–336. [CrossRef]
- 34. Barber, B.M.; Odean, T. The behavior of individual investors. In *Handbook of the Economics of Finance;* Constantinides, G., Harris, M., Stulz, R., Eds.; Elsevier: Amsterdam, The Netherlands, 2011; pp. 1533–1570.

- 35. List, J.A. Does market experience eliminate market anomalies? The case of exogenous market experience. *Am. Econ. Rev.* **2011**, *101*, 313–317. [CrossRef]
- 36. Bradbury, M.A.; Hens, T.; Zeisberger, S. Improving investment decisions with simulated experience. *Rev. Financ.* **2014**, *19*, 1019–1052. [CrossRef]
- Itzkowitz, J.; Itzkowitz, J. Name-Based Behavioral Biases: Are Expert Investors Immune? J. Behav. Financ. 2017, 18, 180–188. [CrossRef]
- 38. Chevalier, J.; Ellison, G. Are some mutual fund managers better than others? Cross-sectional patterns in behavior and performance. *J. Financ.* **1999**, *54*, 875–899. [CrossRef]
- Agarwal, S.; Driscoll, J.C.; Gabaix, X.; Laibson, D. The age of reason: Financial decisions over the lifecycle. In National Bureau of Economic Research Working Paper no. 13191; Harvard University and MIT: Cambridge, MA, USA, 2007.
- 40. Bhandari, G.; Deaves, R. The demographics of overconfidence. J. Behav. Financ. 2006, 7, 5–11. [CrossRef]
- 41. Xiao, W. Does practice make perfect? Evidence from individual investors' experiences and investment returns. *J. Interdiscip. Math.* **2015**, *18*, 811–825. [CrossRef]
- 42. Bodnaruk, A.; Simonov, A. Do financial experts make better investment decisions? *J. Financ. Intermed.* 2015, 24, 514–536. [CrossRef]
- 43. Wulfmeyer, S. Irrational Mutual Fund Managers: Explaining Differences in Their Behavior. *J. Behav. Financ.* **2016**, *17*, 99–123. [CrossRef]
- 44. Chang, C.H. Exploring stock recommenders' behavior and recommendation receivers' sophistication. *J. Econ. Financ.* **2017**, *41*, 1–26. [CrossRef]
- Helbing, D. Agent-based modeling. In *Social Self-Organization-Understanding Complex Systems*; Helbing, D., Ed.; Springer: Berlin/Heidelberg, Germany, 2012; pp. 25–70.
- 46. Wang, L.; Ahn, K.; Kim, C.; Ha, C. Agent-based models in financial market studies. *J. Phys. Conf. Ser.* 2018, 1039, 012022. [CrossRef]
- 47. Munkh-Ulzii, B.; McAleer, M.; Moslehpour, M.; Wong, W.-K. Confucius and herding behaviour in the stock markets in China and Taiwan. *Sustainability* **2018**, *10*, 4413. [CrossRef]
- 48. VanderWeele, T.J.; Shpitser, I. On the definition of a confounder. *Ann. Stat.* **2013**, *41*, 196–220. [CrossRef] [PubMed]
- 49. Bernerth, J.B.; Aguinis, H. A critical review and best-practice recommendations for control variable usage. *Pers. Psychol.* **2016**, *69*, 229–283. [CrossRef]
- 50. Duffy, B.; Smith, K.; Terhanian, G.; Bremer, J. Comparing data from online and face-to-face surveys. *Int. J. Mark. Res.* **2005**, 47, 615–639. [CrossRef]
- 51. Dooley, L.M.; Lindner, J.R. The handling of nonresponse error. *Hum. Resour. Dev. Q.* 2003, 14, 99–110. [CrossRef]
- 52. Podsakoff, P.M.; MacKenzie, S.B.; Podsakoff, N.P. Sources of method bias in social science research and recommendations on how to control it. *Annu. Rev. Psychol.* **2012**, *63*, 539–569. [CrossRef] [PubMed]
- 53. Abreu, M.; Mendes, V. Information, overconfidence and trading: Do the sources of information matter? *J. Econ. Psychol.* **2012**, *33*, 868–881. [CrossRef]
- 54. Mishra, K.; Metilda, M.J. A study on the impact of investment experience, gender, and level of education on overconfidence and self-attribution bias. *IIMB Manag. Rev.* **2015**, *27*, 228–239. [CrossRef]
- 55. Kember, D.; Leung, D.Y.; Jones, A.; Loke, A.Y.; McKay, J.; Sinclair, K.; Tse, H.; Webb, C.; Yuet Wong, F.K.; Wong, M. Development of a questionnaire to measure the level of reflective thinking. *Assess. Eval. High. Educ.* 2000, 25, 381–395. [CrossRef]
- 56. Taylor, E.W. An update of transformative learning theory: A critical review of the empirical research (1999–2005). *Int. J. Lifelong Educ.* **2007**, *26*, 173–191. [CrossRef]
- 57. Waweru, N.M.; Munyoki, E.; Uliana, E. The effects of behavioural factors in investment decision-making: A survey of institutional investors operating at the Nairobi Stock Exchange. *Int. J. Bus. Emerg. Mark.* 2008, *1*, 24–41. [CrossRef]
- Kengatharan, L.; Kengatharan, N. The influence of behavioral factors in making investment decisions and performance: Study on investors of Colombo Stock Exchange, Sri Lanka. *Asian J. Financ. Account.* 2014, 6, 1–23. [CrossRef]
- 59. Fisher, M.; King, J.; Tague, G. Development of a self-directed learning readiness scale for nursing education. *Nurse Educ. Today* **2001**, *21*, 516–525. [CrossRef] [PubMed]

- 60. Fisher, M.; King, J. The self-directed learning readiness scale for nursing education revisited: A confirmatory factor analysis. *Nurse Educ. Today* **2010**, *30*, 44–48. [CrossRef] [PubMed]
- 61. Williams, B.; Brown, T. A confirmatory factor analysis of the self-directed learning readiness scale. *Nurs. Health Sci.* **2013**, *15*, 430–436. [CrossRef] [PubMed]
- Wong, K.K.-K. Mediation analysis, categorical moderation analysis, and higher-order constructs modeling in Partial Least Squares Structural Equation Modeling (PLS-SEM): A B2B Example using SmartPLS. *Mark. Bull.* 2016, 26.
- 63. Kale, P.; Singh, H.; Perlmutter, H. Learning and protection of proprietary assets in strategic alliances: Building relational capital. *Strateg. Manag. J.* **2000**, 217–237. [CrossRef]
- 64. Becker, J.-M.; Rai, A.; Rigdon, E. Predictive validity and formative measurement in structural equation modeling: Embracing practical relevance. In Proceedings of the 34th International Conference on Information Systems, Milan, Italy, 15–18 December 2013.
- 65. Evermann, J.; Tate, M. Assessing the predictive performance of structural equation model estimators. *J. Bus. Res.* **2016**, *69*, 4565–4582. [CrossRef]
- 66. Sarstedt, M.; Ringle, C.M.; Hair, J.F. Partial least squares structural equation modeling. In *Handbook of Market Research*; Springer: Berlin/Heidelberg, Germany, 2017; pp. 1–40.
- 67. Hair, J.F.; Sarstedt, M.; Ringle, C.M.; Gudergan, S.P. *Advanced Issues in Partial Least Squares Structural Equation Modeling*; SAGE Publications: Newbury Park, CA, USA, 2017.
- 68. Sarstedt, M.; Ringle, C.M.; Smith, D.; Reams, R.; Hair, J.F. Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. *J. Fam. Bus. Strategy* **2014**, *5*, 105–115. [CrossRef]
- 69. Geisser, S. A predictive approach to the random effect model. Biometrika 1974, 61, 101–107. [CrossRef]
- Stone, M. Cross-validatory choice and assessment of statistical predictions. J. R. Stat. Soc. Ser. B (Methodol.) 1974, 36, 111–147. [CrossRef]
- 71. Hair, J.F.; Hollingsworth, C.L.; Randolph, A.B.; Chong, A.Y.L. An updated and expanded assessment of PLS-SEM in information systems research. *Ind. Manag. Data Syst.* **2017**, *117*, 442–458. [CrossRef]
- 72. Henseler, J.; Chin, W.W. A comparison of approaches for the analysis of interaction effects between latent variables using partial least squares path modeling. *Struct. Equ. Model.* **2010**, *17*, 82–109. [CrossRef]
- 73. Hulland, J. Use of partial least squares (PLS) in strategic management research: A review of four recent studies. *Strateg. Manag. J.* **1999**, *20*, 195–204. [CrossRef]
- 74. Nunnally, J.C.; Bernstein, I. Psychometric Theory; McGraw-Hill: New York, NY, USA, 1994; Volume 3.
- 75. Gefen, D.; Straub, D.W.; Boudreau, M. Structural Equation Modeling and Regression: Guidelines for Research Practice. *Commun. Assoc. Inf. Syst.* **2000**, *4*, 1–79. [CrossRef]
- 76. Fornell, C.; Larcker, D.F. Structural equation models with unobservable variables and measurement error: Algebra and statistics. *J. Mark. Res.* **1981**, *18*, 382–388. [CrossRef]
- 77. Chin, W.W. The partial least squares approach to structural equation modeling. *Mod. Methods Bus. Res.* **1998**, 295, 295–336.
- 78. Chin, W.W. How to write up and report PLS analyses. In *Handbook of Partial Least Squares*; Springer: Berlin/Heidelberg, Germany, 2010; pp. 655–690.
- 79. Henseler, J.; Ringle, C.M.; Sarstedt, M. A new criterion for assessing discriminant validity in variance-based structural equation modeling. *J. Acad. Mark. Sci.* **2015**, *43*, 115–135. [CrossRef]
- Cassel, C.; Hackl, P.; Westlund, A.H. Robustness of partial least-squares method for estimating latent quality structures. J. Appl. Stat. 1999, 26, 435–446. [CrossRef]
- 81. Hair, J.F.; Ringle, C.M.; Sarstedt, M. PLS-SEM: Indeed a silver bullet. J. Mark. Theory Pract. 2011, 19, 139–152. [CrossRef]
- 82. Bagozzi, R.P.; Yi, Y. On the evaluation of structural equation models. *J. Acad. Mark. Sci.* **1988**, *16*, 74–94. [CrossRef]

84. Cohen, J. *Statistical Power Analysis for the Behavioral Sciences*, 2nd ed.; Lawrence Earlbaum Associates: Mahwah, NJ, USA, 1988.



© 2019 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).